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DAl LY REPO RT Date: October 26, 2007

October 22, 2007- Fair

Maroney/Terpstra Champions meeting held 1000-1130. Touched on SAS issues with coordination and MEP
discussed. Meeting at 1200 to discuss the Hinge A lowering by the Skyway contract. Direction given by Maroney to
only do what can be done without extending the end date of the skyway. Practically this means that no work to
allow lowering will be done beyond modifying the shear restraints. Follow up meeting held at 1445 with TYL on the
phone. Denis and Sagit refused to attend in person, only by phone and were uncooperative.

October 23, 2007 — fair, warm
Attended the OSC BCE meeting in Scotts Valley

October 24, 2007 - fair, warm

Attended Core mtg at 0900. Crane work going well in China and Barge work progressing at OIW. OBG plate
burning started per Flowers. Confirmed that ZPMC will detail tower lifts 3, 4, and 5. Using “TEKLA” software that
Flowers says is cumbersome. Working drawings — al agreed to be 10K to 15K total and that we are about 40%
complete. Dan Raynor to go to Shanghai in January — Who will replace him? Unknown. PE really needed? Ask Lai
Direction to check into painting the template at ZPMC. Marwan raised issue that Concrete and Tower mtgs going
well but OBG and Cable are not. How to Handle?? OBG issues of tolerances and Cable issues on fre hanging
options and miscommunication.

RFI on trolley beam skew mentioned and one on Vortex plates extending through field slices. Questions over
whether there are new changes to the CCO 44 barrier CCO need to check

Detailing by Walter Gatti should be about done in Dec. of this year and will start on the MEP catchup and finl as
built shop drawings. Aslo mentioned were RFI 945 and an RFI on the cable band angles wrt sensitivity of future
changes.

Attended the BCE meeting in Scott's Valley in the PM

October 25, 2007 — fair

Met with Pedro and staff in the AM on E2/T1 issues and updates.
Meeting has been set with TBS and KFM for tomorrow on NOPC 6 issues

Worked on email, PMIV, and TEC through the morning.
Meeting at 1400 to discuss cable issues internally between CT and TYL.

October 26, 2007 — fair
Worked on E2/T1 contract issues all day - NOPC 6 meeting with KFM and Transbay
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